Author's response to reviews

Title: Perioperative complications in infant cleft repair

Authors:

Thomas Fillies (fillies@uni-muenster.de)
Christoph Homann (Christophhomann@aol.com)
Ulrich Meyer (ulrich.meyer@med.uni-duesseldorf.de)
Alexander Reich (reich@anit.uni-muenster.de)
Ulrich Joos (Ulrich.Joos@ukmuenster.de)
Richard Werkmeister (Rwerkmeis@aol.com)

Version: 2 Date: 8 January 2007

Author's response to reviews:

Dear Sir or Madam,

enclosing we send you our revised manuscript "Perioperative complications in infant cleft repair " with comments to revision.

We are looking forward to hearing from you.

Yours faithfully
Thomas Fillies

Comments to revision:

Thank you very much for good suggestions which we included in the revised paper.

In material and methods we added the clinical setting with the anaesthetic protocol.

Units of blood measurements were changed from gms in mls.

In results and in table 1-3 we added the percental statistical analysis.